

Article Unveiling the Power of Stochastic Methods: Advancements in Air Pollution Sensitivity Analysis of the Digital Twin

Venelin Todorov ^{1,2} and Ivan Dimov ^{1,*}

- ¹ Institute of Information and Communication Technologies, Bulgarian Academy of Sciences, Acad. G. Bonchev Str. Bl. 25A, 1113 Sofia, Bulgaria; venelin@parallel.bas.bg or vtodorov@math.bas.bg
- ² Institute of Mathematics and Informatics, Bulgarian Academy of Sciences, Acad. G. Bonchev Str. Bl. 8, 1113 Sofia, Bulgaria
- * Correspondence: ivdimov@bas.bg

Abstract: Thorough examination of various aspects related to the distribution of air pollutants in a specific region and the factors contributing to high concentrations is essential, as these elevated levels can be detrimental. To accomplish this, the development and improvement of a digital twin that encompasses all relevant physical processes in the atmosphere is necessary. This tool, known as DIGITAL AIR, has been created, and it is now necessary to extend it with precise sensitivity analysis. DIGITAL AIR is gaining popularity due to its effectiveness in addressing complex problems that arise in intricate environments; this motivates our further investigations. In this paper, we focus on the preparation and further investigation of DIGITAL AIR through sensitivity analysis with improved stochastic approaches for investigating high-level air pollutants. We discuss and test the utilization of this digital tool in tackling the issue. The unified Danish Eulerian model (UNI-DEM) plays a crucial role within DIGITAL AIR. This mathematical model, UNI-DEM, is highly versatile and can be applied to various studies concerning the adverse effects caused by elevated air pollution levels.

Keywords: air pollution modeling; sensitivity analysis; multidimensional integrals; Monte Carlo methods; digital sequences

MSC: 60J22; 62P12; 65C05; 68W20

1. Introduction

Environmental security is a global priority, and numerous challenges exist in this field. This topic is sensitive and holds great importance for society and the healthcare system. There are several very compelling reasons for this.

The first reason is ecological balance. The environment is a complex web of interconnected ecosystems and species. By safeguarding the environment, we ensure the preservation of biodiversity, which is vital for maintaining ecological balance and the overall health of the planet.

For society, the most important topic is the human health. Environmental degradation, pollution, and climate change have significant implications on human health. Air and water pollution, exposure to harmful chemicals, and climate-related disasters can lead to various health issues, including respiratory diseases, waterborne illnesses, and increased vulnerability to extreme weather events.

Environmental protection is closely linked to sustainable development. By conserving natural resources, promoting renewable energy, and adopting sustainable practices, we can meet the needs of the present generation without compromising the ability of future generations to meet their own needs.

The threat of climate change poses significant challenges to societies worldwide. Addressing climate change requires reducing greenhouse gas emissions, transitioning to cleaner energy



Citation: Todorov, V.; Dimov, I. Unveiling the Power of Stochastic Methods: Advancements in Air Pollution Sensitivity Analysis of the Digital Twin. *Atmosphere* **2023**, *14*, 1078. https://doi.org/10.3390/ atmos14071078

Academic Editors: Miodrag Rancic and Ivana Tosic

Received: 25 May 2023 Revised: 20 June 2023 Accepted: 22 June 2023 Published: 26 June 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). sources, and adapting to the changing climate. Environmental protection plays a crucial role in mitigating the impacts of climate change and ensuring a sustainable future.

Environmental protection can contribute to economic growth and prosperity. The conservation of natural resources, sustainable agriculture, and the development of green technologies create new job opportunities and promote innovation. Additionally, investing in environmental protection can lead to long-term cost savings by reducing environmental risks and the need for expensive environmental remediation.

Environmental issues transcend national boundaries and require global cooperation. Environmental protection provides a common ground for countries to collaborate and work towards shared goals. International agreements and initiatives, such as the Paris Agreement, highlight the importance of collective action in addressing global environmental challenges.

Given these reasons, environmental protection is recognized as a top priority to ensure a sustainable and prosperous future for both current and future generations.

One of the notable findings emphasized in the International Panel for Climate Change's (IPCC) Sixth Report (AR6), published in August 2021, is the consistent warming trend observed over the past four decades. According to the report, each successive decade has been warmer than any preceding decade (Paragraph A.1.2., in [1] (p. 6)). This warming trend has direct implications for the levels of certain air pollutants, including those that can pose risks to plants, animals, and human health. As temperatures continue to rise, the impact of pollution can be significantly intensified.

The IPCC's explicit statement regarding future temperature increases highlights the importance of investigating the influence of climate change on pollution levels. The relationship between temperature and pollutant concentrations carries significant implications, especially considering the anticipated climate changes. Understanding these dynamics is increasingly vital as we strive to comprehend the potential consequences of climate change.

We will utilize a digital twin named DIGITAL AIR, which falls under the increasingly popular trend of digital twin applications [2–6]. DIGITAL AIR comprises a wide range of numerical algorithms, multiple splitting techniques, a plethora of graphical tools, a diverse set of useful scenarios, extensive meteorological and emission data files, and a comprehensive repository of geographical information. This includes detailed information about numerous cities in Europe and the borders of European countries. The study is employing several tools: the UNI-DEM mathematical model, which necessitates the implementation of efficient and accurate numerical algorithms on modern supercomputers; extensive datasets comprising meteorological, emission, and geographical information; carefully designed climatic scenarios that account for future temperature increases and the corresponding rise in natural (biogenic) emissions; and graphical programs for visualizing the numerical results obtained. More importantly, we introduce further investigation of DIGITAL AIR by conducting a sensitivity analysis using advanced stochastic methods.

We introduce an enhanced version of the lattice sequence with product- and orderdependent weights, which demonstrates some improvements compared to the best available stochastic sequences used to measure sensitivity indices in the digital ecosystem under investigation. We conduct a comprehensive comparison with the best available modifications of the Sobol sequences for multidimensional sensitivity analysis. This analysis aims to explore the model's output concerning variations in input emissions of anthropogenic pollutants and assess the rates of various chemical reactions.

A brief overview of the structure of the primary tool utilized in DIGITAL AIR, which is the large-scale air pollution model UNI-DEM, is given in [7]. For more extensive details about UNI-DEM and the various numerical procedures employed in its treatment, readers can refer to [8–11]. Moreover, other publications such as [12–14] discuss different applications of this model. We will delve into the main principles underlying the climatic scenarios implemented in DIGITAL AIR. These principles align with those employed in several prior papers, as outlined in [15–19]. However, we have also taken into consideration recommendations proposed in [1]. To gain a more precise understanding of the study's purpose, it is advisable to refer to the following sources as well [20–22]. By utilizing DIGITAL AIR, the assessment of ozone levels extends beyond the context of Bulgaria and encompasses other European countries as well. Notably, findings for Denmark, Hungary, and countries within the Balkan Peninsula were presented in references [14,15,19]. DIGITAL AIR has the capability to examine additional hazardous pollution levels, such as those resulting from emissions of sulfur dioxide (SO₂) and nitrogen oxides (NO_x), as outlined in [18].

The paper is organized as follows: Section 2 provides the key definitions of sensitivity analysis and Sobol sensitivity indices. Section 3 offers a concise overview of the structure of UNI-DEM, the primary air pollution model utilized in DIGITAL AIR, along with its main climatic scenarios. Section 4 presents the preliminary calculations conducted with the mathematical model and outlines the methodology for approximation stage before calculating the sensitivity indices. In Section 5, a brief description of the stochastic algorithms based on Sobol and lattice sequences is provided. Section 6 presents the numerical results obtained from employing advanced stochastic approaches to evaluate the sensitivity indices. Section 7 presents discussion about the obtained results. Finally, Section 8 concludes the paper with some closing remarks.

2. Sensitivity Analysis

Sensitivity analysis (SA) [23–28] is a technique used to assess the impact of changes in input variables on the output or outcome of a mathematical or computational model. It involves systematically varying the values of input parameters within a specified range and examining how these variations affect the model's results. The goal of SA is to understand the relative importance and influence of different input factors or variables on the model's output. By analyzing sensitivity, researchers can identify which variables have the most significant impact on the model's behavior and outcomes, allowing for a better understanding of the system being studied. SA is widely used in various fields, including engineering, finance, environmental modeling, and decision-making processes, to gain insights into the robustness, reliability, and sensitivity of models and their outputs.

SA is a contemporary and promising approach utilized in the investigation of extensive systems, including ecological systems, as documented in references [25,29,30]. This technique revolves around the estimation or prediction of a metric quantifying the responsiveness of model outputs to variations in input parameters through extensive computer simulations on complex mathematical models. Mathematically, this problem is formulated as a set of integrals with high dimensions.

Efficient Monte Carlo (MC and quasi-Monte Carlo (QMC) methods [31–33] play a crucial role in conducting SA for large-scale computer models, ensuring optimal utilization of computational resources. These methods prove particularly valuable for analyzing intricate models characterized by a multitude of input parameters, as they can handle substantial volumes of data and yield rapid and accurate outcomes.

The rigid definition of SA in the book of Saltelli et al. [34] is the following.

Definition 1. *SA is the study of how the variation in the output values of a model can be distributed among the different sources of variation among the input parameters." SA consists of the following stages* [24,25]:

- define the model and its input and output parameters,
- determine the corresponding density functions for each input parameter,
- generate the so-called "input matrix" of values with an appropriate random sample generation method,
- compute the model output based on the generated values,
- analyze the fluctuations in the model output,
- *estimate the influence (or relative importance) of the input parameters on the fluctuations in the output.*

There are various approaches available for conducting sensitivity analysis (SA), as referenced in [29]. The choice of SA method depends on the behavior of the model,

including factors such as linearity, monotonicity, and additivity in the relationship between input parameters and output.

SA can be divided into two distinct classes: local and global. Local SA focuses on understanding how small changes in values around a fixed point impact the variability of output values. This method involves investigating the sensitivity of individual parameters while keeping other parameters constant and only allowing the selected parameter to vary. It is commonly known as the "one at a time" (OAT) technique. However, OAT experiments do not consider the joint influence of multiple parameters. On the other hand, global SA examines the entire range of variation in input parameter values. Parameter screening methods, such as the ones discussed in [35], are specifically designed for models with a large number of input parameters or those that pose estimation challenges. Although these methods are useful for sensitivity analysis, they have a key limitation. They provide a qualitative estimation of sensitivity by grouping input parameters based on their influence, but they do not provide a quantitative measure of the individual importance of each parameter in relation to the others.

Variance-based methods (VBM) are widely used quantitative techniques for SA [25]. These methods employ random samples and are particularly suitable for MC simulations. Input parameters are modeled as random variables characterized by probability density functions. The main objective of VBM is to identify the input parameters that have the most significant impact on output variations and determine which parameters require more accurate estimation to reduce uncertainties in output values.

When conducting a detailed analysis of concentration sensitivities in large mathematical models, it is beneficial to introduce stochastic variables and equations. Sobol's work in [36] provides a successful systematic framework for this theory.

Let the mathematical model be represented by a model function

$$u = f(x)$$
, where $x = (x_1, x_2, \dots, x_d) \in U^d \equiv [0; 1]^d$ (1)

is a vector of input parameters with joint probability density $p(x) = p(x_1, ..., x_d)$. We assume that the output is a scalar and that the input parameters are independent and the probability density $p(x) = p(x_1, x_2, ..., x_d)$ is known. Therefore, the output parameter u is a random variable as a function of the random vector x.

We now introduce the measure of the degree of influence of an input parameter on the output.

Definition 2 (*First-order sensitivity index* [37]). The basic indicator corresponding to a given input parameter x_i , i = 1, ..., d (normalized between 0 and 1) is called the Sobol first-order sensitivity index (SI) [37] and is defined as follows

$$\frac{\mathbf{D}[\mathbf{E}[\mathbf{u}|x_i]]}{\mathbf{D}_{\mathbf{u}}},\tag{2}$$

where $\mathbf{D}[\mathbf{E}[\mathbf{u}|x_i]]$ is the variance of the conditional mathematical expectation of \mathbf{u} on x_i , and $\mathbf{D}_{\mathbf{u}}$ is the full variance on \mathbf{u} .

We now give a definition of the total SI.

Definition 3. Total sensitivity index (TSI) [27] is a measure of the overall influence (full effect) of an input parameter on variations in the output. The TSI of the input parameter $x_i, x_i \in 1, ..., d$ is defined as follows [27,37]:

$$S_{x_i}^{tot} = S_i + \sum_{l_1 \neq i} S_{il_1} + \sum_{l_1, l_2 \neq i, l_1 < l_2} S_{il_1 l_2} + \dots + S_{il_1 \dots l_{d-1}},$$
(3)

where S_i is called the main effect (first-order SI) of x_i and $S_{il_1...l_{j-1}}$ is the *j*-th-order SI (two-way interactions for j = 2 and three-way interactions for j = 3, etc.) for the parameter x_i ($2 \le j \le d$).

The degree of joint influence of the input parameters $x_{i_1}, \ldots, x_{i_v}, v \in \{2, \ldots, d\}$ on the variation of the result is described by the higher-order addends. In [38], it is shown that small subsets of the input parameters will contribute significantly to the output in multivariate models. Therefore, higher-dimensional adders can be neglected and lower-order SIs can be used, but the contribution of higher-order adders can be controlled.

The set of input parameters is classified depending on the TSI as $S_{x_i}^{tot}$ [24]: extremely important if $0.8 < S_{x_i}^{tot}$, important if $0.5 < S_{x_i}^{tot} < 0.8$, unimportant if $0.3 < S_{x_i}^{tot} < 0.5$, and insignificant if $S_{x_i}^{tot} < 0.3$.

We now consider one of the most commonly used VBMs, namely Sobol's [36,37] method for computing global sensitivity indices (GSIs). The main advantage of this method is it computes not only first-order SIs but also higher-order SIs, and TSI can compute with only one integral (one subintegral function) for a given parameter by the MC method.

Sobol's method for global SA relies on a decomposing the integrable model function f (in *s*-dimensional input parameter space) into addends of increasing dimensionality.

Definition 4. The high-dimensional model representation (HDMR representation) is defined as

$$f(\mathbf{x}) = f_0 + \sum_{\nu=1}^d \sum_{l_1 < \dots < l_\nu} f_{l_1 \dots l_\nu}(x_{l_1}, x_{l_2}, \dots, x_{l_\nu}),$$
(4)

where f_0 is a constant. The total number of addends in Equation (4) is 2^d (see [39]).

In general, this representation is not unique to [37]. The key feature of (1) is that it can be represented by small subsets of input parameters [38,40], and this assumption is the basis of (4). Thus, functions on more variables describing the effect of the interaction of input parameters in (4) can be neglected. This will reduce the dimensionality of the problem.

Definition 5. The representation (4) is unique and is called an ANOVA representation (analysis of variance, analysis of variance) of the model function f(x) [41], under the condition that for each additive is valid

$$\int_0^1 f_{l_1...l_\nu}(x_{l_1}, x_{l_2}, \dots, x_{l_\nu}) dx_{l_k} = 0, \quad 1 \le k \le \nu, \quad \nu = 1, \dots, d.$$
(5)

Sobol has proven [36] that the decomposition (4) is if and only if (5) holds and the functions on the right-hand side can be represented in a unique way [41]:

• $f_0 = \int_{U^d} f(\mathbf{x}) d\mathbf{x};$ • $f_{l_1}(x_{l_1}) = \int_{U^{d-1}} f(\mathbf{x}) \prod_{k \neq l_1} d\mathbf{x}_k - f_0, \ l_1 \in \{1, 2, \dots, d\};$

•
$$f_{l_1 l_2}(x_{l_1}, x_{l_2}) = \int_{U^{d-2}} f(\mathbf{x}) \prod_{k \neq l_1, l_2} d\mathbf{x}_k - f_0 - f_{l_1}(x_{l_1}) - f_{l_2}(x_{l_2})$$

 $l_1, l_2 \in \{1, 2, \dots, d\}.$

Because the above subsets of indices differ from each other by at least one element and the corresponding integral is equal to zero for the corresponding index by applying (5), it follows that the addends in the ANOVA representation are mutually orthogonal

$$\int_{U^d} f_{i_1...i_{\mu}} f_{j_1...j_{\nu}} \, \mathrm{d} \mathbf{x} = 0, \quad (i_1, \ldots, i_{\mu}) \neq (j_1, \ldots, j_{\nu}), \quad \mu, \nu \in \{1, \ldots, d\}.$$

Definition 6. *The quantities*

$$\mathbf{D} = \int_{\mathcal{U}^d} f^2(\mathbf{x}) d\mathbf{x} - f_0^2, \quad \mathbf{D}_{l_1 \dots l_\nu} = \int f_{l_1 \dots l_\nu}^2 dx_{l_1} \dots dx_{l_\nu}$$
(6)

are called variances (total and partial variances, respectively) and are obtained after squaring and integrating over U^d the Equation (4) under the assumption that f(x) has a summable square.

Thus, we arrive at the following definition:

Definition 7. *The total variance of the model output parameter is decomposed into the partial variances* [36] *in an analogous way to the decomposition of the model function, which is only an ANOVA-type representation:*

$$\mathbf{D} = \sum_{\nu=1}^{d} \sum_{l_1 < \dots < l_{\nu}} \mathbf{D}_{l_1 \dots l_{\nu}}.$$
(7)

It will now be shown how the corresponding SI $S_{l_1 \dots l_{\nu}}$ are defined by the conditional expectation variances $\mathbf{D}_{l_1} = \mathbf{D}[f_{l_1}(x_{l_1})] = \mathbf{D}[\mathbf{E}(\mathbf{u}|x_{l_1})], \mathbf{D}_{l_1 \dots l_{\nu}}, 2 \leq \nu \leq d$ (see Equation (8)).

Definition 8 ([36,41]). *The quantities*

$$S_{l_1\dots l_{nu}} = \frac{\mathbf{D}_{l_1\dots l_{nu}}}{\mathbf{D}}, \quad \nu \in \{1,\dots,d\}$$
(8)

are called Sobol GSIs.

This formula coincides with (2) for $\nu = 1$, and so, the measures defined correspond to the main effect and the interaction effect between the parameters. Now, dividing (7) by **D**, it follows that the following properties of these indices hold:

$$S_{l_1 \dots l_{\nu}} \ge 0,$$
 and $\sum_{\nu=1}^d \sum_{l_1 < \dots < l_{\nu}}^d S_{l_1 \dots l_{\nu}} = 1.$

Definition 9. Let us have a set of m variables $(1 \le m \le d - 1)$:

$$y = (x_{k_1}, \ldots, x_{k_m}), \qquad 1 \le k_1 < \ldots < k_m \le d,$$

and let z be the set of the remaining d - m variables, $K = (k_1, ..., k_m)$, and x = (y, z). Then, the variances corresponding to the sets of variables y and z are defined as

$$\mathbf{D}_{y} = \sum_{n=1}^{m} \sum_{(i_{1} < \dots < i_{n}) \in K} \mathbf{D}_{i_{1} \dots i_{n}}, \qquad \mathbf{D}_{z} = \sum_{n=1}^{d-m} \sum_{(j_{1} < \dots < j_{n}) \in \bar{K}} \mathbf{D}_{j_{1} \dots j_{n}},$$
(9)

where the complement of the subset K to the set of indices of all input parameters is denoted by \bar{K} , the first sum in (9) is over all subsets (i_1, \ldots, i_n) , where all indices i_1, \ldots, i_n belong to K and the full variance of the subset $y - \mathbf{D}_y^{tot} = \mathbf{D} - \mathbf{D}_z$ is over all subsets (i_1, \ldots, i_v) , $1 \le v \le d$, where at least one index belongs to K: $i_l \in K$, $1 \le l \le v$.

The procedure for calculating the GSI is based on the following representation of the variance \mathbf{D}_{y} : $\mathbf{D}_{y} = \int f(x) f(y, z') dx dz' - f_{0}^{2}$ (see [41]). The last equality allows for the construction of an MC algorithm to compute f_{0} , \mathbf{D} and \mathbf{D}_{y} , where $\xi = (\eta, \zeta)$:

$$\frac{1}{N} \sum_{j=1}^{N} f(\xi_j) \xrightarrow{P} f_0, \qquad \qquad \frac{1}{N} \sum_{j=1}^{N} f(\xi_j) f(\eta_j, \zeta_j') \xrightarrow{P} \mathbf{D}_{\mathbf{y}} + f_0^2, \\ \frac{1}{N} \sum_{j=1}^{N} f^2(\xi_j) \xrightarrow{P} \mathbf{D} + f_0^2, \qquad \qquad \frac{1}{N} \sum_{j=1}^{N} f(\xi_j) f(\eta_j', \zeta_j) \xrightarrow{P} \mathbf{D}_{\mathbf{z}} + f_0^2.$$

Therefore, for $m = 1, y = \{x_{l_1}\}, l_1 \in \{1, ..., d\}$ and $z = \{1, ..., d\} \setminus l_1$:

$$S_{l_1} = S_{(l_1)} = rac{\mathbf{D}_{(l_1)}}{\mathbf{D}}, \quad S_{l_1}^{tot} = rac{\mathbf{D}_{l_1}^{tot}}{\mathbf{D}} = 1 - S_z.$$

Following the idea of Homa and Saltelli in [27], a better first-order approximation of SI is obtained if f_0^2 in (6) is approximated directly using

$$f_0^2 = \int_{U^{2d}} f(\mathbf{x}) f(\mathbf{x}') d\mathbf{x} d\mathbf{x}'$$

instead of $f_0^2 = \left[\int_{U^d} f(\mathbf{x}) d\mathbf{x}\right]^2$. This follows from the Formula (6), where the corresponding estimate of the first-order indices $S_{l_1}, l_1 \in \{1, ..., d\}$ tends to zero for the corresponding input parameter x_{l_1} if (2) [30] is used and can be obtained from the formula:

$$D_{l_1} = \int_{U^{2d}} f(\mathbf{x}) \left[f(x'_1, \dots, x'_{l_1-1}, x_{l_1}, x'_{l_1+1}, \dots, x'_d) - f(x'_1, \dots, x'_d) \right] d\mathbf{x} d\mathbf{x}'.$$

Saltelli's idea in [30] for computing TSI is to use the following estimate for $S_{l_1}^{tot}$:

$$1 - \frac{1}{\mathbf{D}} \left\{ \int_{U^{d+1}} f(x_1, \dots, x_d) f(x_1, \dots, x_{l_1-1}, x'_{l_1}, x_{l_1+1}, \dots, x_d) d\mathbf{x} d\mathbf{x}'_{l_1} - f_0^2 \right\}.$$

The computational complexity of the aforementioned method for computing all firstorder SIs and all TSIs is determined by the number of function evaluations required, which is proportional to N(2d + 1) (N values of the function for f_0 , dN values of the SIs function, and dN values of the TSIs function), where N represents the sample size and d is the number of dimensions.

In contrast, commonly used VBMs such as Sobol's method and FAST have a computational complexity that scales linearly with dN when estimating all first-order SIs and TSIs for the input parameters (see [30]).

This illustrates that the core of the SA problem resides on the computation of TSIs (3) and, more specifically, the Sobol GSIs of the corresponding order (8). This computation can be reduced to the evaluation of multidimensional integrals:

$$I = \int_{\Omega} g(\mathbf{x}) p(\mathbf{x}) \, \mathrm{d}\mathbf{x}, \ \Omega \subset \mathbb{R}^d$$

where g(x) is a summable square function in Ω and $p(x) \ge 0$ is a probability density, such that $\int_{\Omega} p(x) dx = 1$.

Consequently, we observe that the computation of the 2^d integrals of the form (6) is necessary to obtain the TSI $S_{x_i}^{tot}$ for a fixed parameter.

The whole methodology for performing SA is given on Figure 1.



Figure 1. Methodology for performing sensitivity analysis.

3. Mathematical Model UNI-DEM

Ongoing research and computational experiments have been conducted utilizing the unified Danish Euler model (UNI-DEM), also known as UNI-DEM, which has proven to be a robust mathematical framework for accurately capturing the relevant physical and chemical processes [8–10,42]. The integration of the unified Danish Eulerian model (UNI-DEM) with various suitable climatic scenarios holds a pivotal and highly significant position within the framework of DIGITAL AIR. Developed by Prof. Zahari Zlatev and their colleagues at the Danish National Institute for Environmental Research [9], UNI-DEM possesses the ability to effectively calculate the concentrations of various hazardous pollutants. It has been widely employed for over two decades in interdisciplinary research and long-term simulations addressing air pollution. Importantly, the proposed SA methodology can be readily extended to other models, showcasing its versatility and applicability beyond UNI-DEM.

UNI-DEM serves as a simulation tool for studying the long-range transport of air pollutants, their temporal changes resulting from chemical and photochemical reactions, and their interactions with the environment. It incorporates crucial physical processes such as advection, diffusion, deposition, emissions, and chemical transformations. The model allows for the analysis of pollutant concentrations over time, specifically focusing on sulfur, nitrogen, ammonia, ammonium ions, nitrogen radicals, and hydrocarbons, which have significant implications for environmental, agricultural, and public health concerns. The geographic scope of the model encompasses Europe, the Mediterranean, and parts of Asia and Africa, with an approximate area coverage of 4800 × 4800 km.

To effectively handle the complexity of the model, it is divided into three subsystems or submodels, each targeting specific physical and chemical processes. By discretizing these submodels and utilizing parallel computing on powerful supercomputers, the model can be executed efficiently in real-time, enabling practical problem-solving within reasonable timeframes [8,9].

Chemical reactions play a pivotal role in the model [43]. The equations within the model accurately represent the system by accounting for chemical reactions. The presence of these reactions contributes to the nonlinearity and "rigidity" of the equation system [43].

The model employs the compressed CBM-IV (carbon bond mechanism) chemical scheme, which has been enhanced in [8]. It encompasses 35 pollutants and 116 chemical reactions, with 69 reactions being time-dependent and 47 being time-independent. This chemical scheme is well-suited for investigating scenarios involving high pollutant concentrations.

Among the model components, chemical reactions are the most challenging and time-consuming, with 69 time-dependent and 47 time-independent reactions requiring careful consideration.

The rate constant in chemical reactions signifies the reaction rate when reactant concentrations are at 1 mol/L, as described by the law of mass action discovered by Guldberg and Waage [30]. Therefore, the intensity of the chemical rate constant directly influences the rate of chemical processes.

The model is described mathematically by [8,10,31] through the following system of partial differential equations:

$$\frac{\partial c_s}{\partial t} = -\frac{\partial (uc_s)}{\partial x} - \frac{\partial (vc_s)}{\partial y} - \frac{\partial (wc_s)}{\partial z} +
+ \frac{\partial}{\partial x} \left(K_x \frac{\partial c_s}{\partial x} \right) + \frac{\partial}{\partial y} \left(K_y \frac{\partial c_s}{\partial y} \right) + \frac{\partial}{\partial z} \left(K_z \frac{\partial c_s}{\partial z} \right) +
+ E_s + Q_s (c_1, c_2, \dots, c_q) - (k_{1s} + k_{2s}) c_s, \quad s = 1, 2, \dots, q.$$
(10)

The number q of equations in (10) is equal to the number of pollutants that are studied by the model. The other quantities included in the model are described below:

 c_s —the pollutant concentrations,

u, *v*, *w*—the wind components along the coordinate axes,

 K_x, K_y, K_z —diffusion coefficients,

 E_s —the emission in the spatial domain,

 k_{1s}, k_{2s} —the dry and wet deposition coefficients, respectively, (s = 1, ..., q),

 $Q_s(c_1, c_2, \ldots, c_q)$ —nonlinear functions describing chemical reactions between pollutants.

The region of study and the computational domain is shown on Figure 2.

The UNI-DEM spatial domain, utilizing a stereographic geographic projection, comprises a surface plane that measures (4800 km \times 4800 km). This plane encompasses Europe and its surrounding regions. Each of the ten horizontal levels is discretized using a grid with dimensions of (10 km \times 10 km). This discretization ensures an ample number of cells, accommodating even small European countries such as Denmark and Bulgaria [10].

The same spatial domain was discretized into a 480×480 grid with a resolution of 10×10 km. Although this refinement significantly increases computational requirements, the comparison between results obtained on the coarse and fine grids demonstrates the value of these efforts, particularly when utilizing the 3-D versions with 10 non-equidistant layers in the vertical direction. However, it was not feasible to acquire input data, including emission and meteorological data, at this high-resolution grid. Instead, the available emission data on the 50 km grid were evenly distributed across 25 smaller grid squares obtained during the transition to the 10 km resolution. To prepare meteorological data for the fine grid, simple linear interpolation is employed, both spatially and temporally.

At the surface level, there are a total of 230, 400 grid squares, each measuring ($10 \text{ km} \times 10 \text{ km}$). The model typically runs with a time step of 30 s, spanning a continuous period of 16 years [31].

Several conditions specified in [8] are assumed. Firstly, the spatial derivatives in the system of PDEs (10) are directly discretized. Secondly, the first-order backward differentiation formula is applied to solve the resulting system of ODEs that arise from the discretization of spatial derivatives. Thirdly, the chosen chemical scheme is the CBM-IV scheme, involving 56 chemical species. Finally, the model is executed for a duration of 16 years.



Figure 2. The region of study and the computational domain.

Under these assumptions, each time step requires the processing of $3600/30 \times 24 \times 365 \times 16 = 16,819,200$ sets of nonlinear algebraic equations. Each of these sets contains $480 \times 480 \times 10 \times 56 = 129,024,000$ equations. To solve these nonlinear algebraic equations, iterative methods are employed, resulting in the solution of very large systems of linear algebraic equations within an inner loop at each time step. The number of these systems during a one-year loop is estimated to be substantial, approximately $O(10^9)$ or even higher.

The UNI-DEM, within the context of DIGITAL AIR, was executed using a total of 14 distinct scenarios spanning a continuous period of 16 years from 1989 to 2004. These scenarios encompass various conditions and factors. The first among the selected five scenarios served as the baseline, providing a reference point for comparison. The subsequent three scenarios were constructed based on assumptions regarding future temperature increases, derived from the conclusions drawn in the IPCC report. The fifth scenario incorporated an additional aspect, considering the anticipated rise in natural emissions of certain air pollutants. Figure 3 illustrates the capability of UNI-DEM to generate dependable outcomes even for the smaller countries in Europe.



Figure 3. Mean monthly concentrations of ozone levels in different European countries.

The baseline scenario uses actual meteorological data and actual emissions data in Europe and its surroundings over the selected period of 16 consecutive years (1989 to 2004), whose data are obtained from the EMEP (European Monitoring and Evaluation Programme) database (for detailed information on the Basic scenarios and other scenarios in the digital twin, see [7]). The definitions presented in [7] outline the First Climate Scenario, which focuses solely on changes in temperature. To visualize the anticipated temperature patterns in Europe, the scenario employs annual temperature changes recommended in various IPCC specialist reports [44]. These changes are used to create a map representing future temperature expectations for the first horizontal level of the digital twin's spatial domain, known as UNI-DEM. This level consists of a grid with dimensions of (480 × 480) cells. The study in [7] reveals that the mean annual temperature change within each cell of the first horizontal level of UNI-DEM corresponds to the prescribed values from the IPCC reports for each of the selected 16 years. The approach adopted in [7] assumes that the expected temperature increase in a particular cell during a given hour between 1989 and 2004 falls within the range of [a, b]. It is demonstrated that the temperature in this cell at a specific hour experiences an increase in $a + \gamma(n)$, where $\gamma(n)$ is a randomly generated and uniformly distributed quantity in the interval [0, b - a]. Consequently, the mathematical expectation of the average annual temperature increase in any cell within the first level of the spatial domain, across any year within the 16-year interval, is equal to (b - a)/2.

Based on the conclusions derived from the IPCC reports, it is projected that extreme events will intensify in the future. Specifically, the Second Climate Scenario, which was analyzed, indicates that maximum daily temperatures will rise, leading to an increased frequency of hot days in terrestrial areas. Additionally, a majority of land regions will experience elevated minimum temperatures, fewer occurrences of cold days, and a decrease in frosty days. Moreover, the diurnal temperature range will shrink in terrestrial regions. These anticipated changes have been taken into consideration in the Second Climate Scenario, and although it incorporates temperature variations similar to those in the First Climate Scenario, it introduces two additional modifications. Firstly, nighttime temperatures are increased by a greater proportion in comparison to daytime temperatures. Secondly, during summer periods, hotter days experience a larger temperature increase.

The Third Climate Scenario, which is the most advanced of the three scenarios, incorporates further findings from the IPCC experts. This scenario expands upon the Second Climate Scenario by considering the following conclusions: increased winter precipitation across land and water, reduced precipitation in continental Europe, adjustments in humidity data, a 10% increase in winter cloud cover, and maintaining the same cloud cover as the Second Climate Scenario during summer. The expected average annual temperature changes remain unchanged. Notably, the Third Climate Scenario is the only one visualized in [7].

The significance of natural (biological) emissions is progressively growing and emerging as a crucial factor. There are at least two underlying factors contributing to this rise. Firstly, there has been a continuous reduction in human-made (anthropogenic) emissions in numerous European countries over the past two to three decades. Secondly, anticipated climatic changes and elevated temperatures are expected to stimulate an increase in natural (biological) emissions. Consequently, it is valuable to develop and implement scenarios incorporating higher natural (biological) emissions. Several scenarios incorporating recommended adjustments to the magnitude of biological (natural) emissions, are employed to address this objective in [7,45,46]. Anticipated temperature rises within the initial horizontal level of the UNI-DEM spatial domain as indicated by the findings of the IPCC reports are shown on Figure 4.

The subsequent analysis will delve into the outcomes obtained across the entire domain of UNI-DEM using the climatic scenarios discussed earlier. Our primary focus will be on evaluating the ozone levels not only throughout Europe but also in specific cities within the region. Of particular concern are instances of high ozone concentrations, as they can have detrimental effects, especially on vulnerable groups such as individuals with respiratory conditions such as asthma. Therefore, we will present detailed findings regarding the extent of these concentrations in various parts of Europe. Specifically, our investigation centers around identifying the occurrence of "bad days". To qualify as a "bad day", we examine the maximum value, denoted as c_max , of the 8 h average ozone concentrations at a given location in Europe on any given day. If $c_m ax$ exceeds 60 ppb at least once during that day, it is categorized as a "bad day". It is imperative to ensure that the number of "bad days" remains within acceptable limits, preferably not exceeding 25 per year as recommended in the Ozone Directive issued by the EU Parliament in 2002.



Figure 4. Anticipated temperature rises within the initial horizontal level of the UNI-DEM spatial domain, as indicated by the findings of the IPCC reports.

Figure 5 provides visual representations illustrating the distribution of "bad days" throughout Europe. The distribution and frequency of "bad days" in different regions of Europe exhibit significant variability from year to year, as evident in the two left-hand side plots, which depict results from the Basic Scenario for the years 1994 and 2004. Implementation of the Third Climatic Scenario generally leads to an increase in the occurrence of "bad days." The magnitude of these changes can be substantial, as indicated in the plots on the right, which present the percentage increases in the number of "bad days" for the selected years. Across numerous parts of Europe, the number of "bad days" exceeds the recommended limit of 25 days, demonstrating a considerable level of exceedance.

Figure 6 presents certain outcomes obtained in the surface domain of UNI-DEM by utilizing the impact of natural (biogenic) emissions on ozone levels in Europe. The two plots reveal significant variations across different regions of Europe and from one year to another. These changes can be substantial and exhibit distinct patterns. Overall, there is a consistent trend observed wherein the increase in biogenic (natural) emissions results in a substantial rise in the frequency of "bad days" across numerous parts of Europe.

Numbers of days with high ozone levels when the Basic Scenario is used



Numbers of days with high ozone levels when the Basic Scenario is used





Figure 5. In the top left graph, the number of "bad days" occurring in 2004 is presented, whereas the bottom-left graph shows the corresponding count of "bad days" in 1994. On the right-hand side, the two accompanying graphs illustrate the percentage increases in the occurrence of "bad days" when comparing the Third Climate Scenario with the respective base years.







Figure 6. The left-hand-side plot of the graph displays the percentage increases in the number of "bad days" in the year 2004, whereas the right-hand-side plot shows the corresponding percentage increases for the year 1994. These increases are observed when utilizing the scenario incorporating elevated natural (biogenic) emissions.

4. Preliminary Calculations with UNI-DEM

By definition, SA includes models, input parameters, and output parameters. In this study, anthropogenic emissions and chemical reaction rate constants are considered as input parameters and pollutant concentrations as output parameters. Mathematically, the input parameters are treated as normally distributed random variables (which is established in [10]) whose mathematical expectation is 1.0. The spatial domain of UNI-DEM is discretized by $96 \times 96 \times 10$ nodes in the three-dimensional version of UNI-DEM.

UNI-DEM experiments were conducted for the period 1994–1998. It is important to pay attention to the fact that a specific year is less important in climate research, because it takes about 30 years to change the climate scenario. The season of the year and the region for which the corresponding climate study was made are much more important. Therefore, a relatively long period in the past was chosen to allow us to compare the results for the data produced by the digital twin with the actual measured pollution data in that period. Furthermore, this comparison shows a high (and previously estimated precision of the digital twin) precision.

The (10) was considered leaving only the adders describing the emissions and chemical reactions. As they do not depend on spatial variables, (10) is reduced to a system of ODEs

$$dg_{s,i,j,k}/dt = E_{s,i,j,k} + Q_{s,i,j,k}(g_{1,i,j,k}, g_{2,i,j,k}, \dots, g_{q,i,j,k}),$$
(11)

where $g_{s,i,j,k}(t)$ is the concentration value c_s at the point (x_i, y_j, z_k) of the grid at time t.

The different stages and components of the SA scheme for UNI-DEM (SA-DEM) is given in Figure 7.



Figure 7. Implementation of the SA scheme for UNI-DEM

The UNI-DEM model is employed at a fixed location within the mesh, and a smaller system known as the "box model" [10] is utilized for sensitivity analysis purposes. The box model represents a reduced system that can be solved repeatedly without computational difficulties, unlike the entire model, which involves solving large systems of ODEs at each time step and can be computationally challenging due to its size (containing millions of equations). In the sensitivity studies, the box model is solved multiple times while varying the rate constants of chemical reactions using a perturbation parameter α , where $\alpha \in 0.1, 0.2, \ldots, 2.0$. Through this computational procedure, it was observed that the concentrations of pollutants are primarily sensitive to changes in the rate constants of the third and twenty-second time-dependent reactions, as well as a sixth time-independent reaction. These initial findings guided the subsequent comprehensive SA employing the aforementioned approaches to obtain more accurate and precise results.

In previous studies [47,48], the identification of critical rate constants of chemical reactions based on a specific criterion was performed. Ozone, known as a highly hazardous air pollutant, was the focus of investigation. The analysis focused on the average concentrations of ozone (O_3) during the summer month of July because it is recognized as the period with the highest ozone concentration.

By iteratively solving the system defined by (11) while varying the rate constants of chemical reactions using a perturbation parameter α ranging from 0.1 to 2.0, the most influential rate constants were determined. Notably, the rate constant of the 22nd timedependent chemical reaction was found to exert the most significant impact on the concentration of ozone (O_3). The effects of this specific rate constant on the concentrations of other pollutants, such as nitrogen dioxide (NO_2), sulfur dioxide (SO_2), peroxide radicals (*PHO*), and nitric oxide (NO), are illustrated in Figure 8. However, the influence on sulfur dioxide concentrations was practically negligible.

Furthermore, it was observed that the influence of the rate constant from the CMB IV chemical reaction on ozone concentration remained relatively consistent across different years. In other words, the pattern of concentration change in response to variations in the perturbation parameter showed a similar trend over time. This behavior is demonstrated in Figure 9.



Figure 8. Sensitivity of several species to change of chemical rates



Figure 9. Sensitivity of ozone concentrations to change of chemical rates.

It is also found that the concentrations of O_3 are most significantly affected by the following chemical reactions: #1, #3, #7 (time-dependent) and #27, #28 (time-independent) reactions of CBM-IV ([8]). The simplified chemical equations of these reactions are as follows:

| [#1] | $NO_2 + h\nu \Longrightarrow NO + O;$ |
|-------|--|
| [#3] | $O_3 + NO \Longrightarrow NO_2;$ |
| [#7] | $NO_2 + O_3 \Longrightarrow NO_3;$ |
| [#22] | $HO_2 + NO \Longrightarrow OH + NO_2;$ |
| [#27] | $HO_2 + HO_2 \Longrightarrow H_2O_2;$ |
| [#28] | $OH + CO \Longrightarrow HO_2$. |

Not all reactions involve ozone; instead, significant ozone precursors are involved. The UNI-DEM calculations primarily focus on obtaining the monthly average concentrations of various hazardous chemical species or groups of species. These concentrations are determined based on the specific chemical scheme and are calculated at grid points within the designated area. The input parameters in focus are the chemical reaction rates, whereas the output parameters of interest are the concentrations of pollutants.

To perform the UNI-DEM calculations, a set of perturbation parameters $\alpha = (\alpha_1, ..., \alpha_6)$ is used within a six-dimensional hypercube ranging from 0.6 to 1.4. The values of α are chosen along the edges of the hypercube, starting from the vertex representing the Basic Scenario with true emissions and extending to all other vertices. Along each edge, the α samples are uniformly distributed by decrementing all variable coordinates by a fixed step of 0.1.

The generated data represent relationships of the form:

$$r_{s}(\alpha) = \frac{c_{s}^{\alpha}(a_{s}^{i_{max}}, b_{s}^{j_{max}})}{c_{s}^{max}}, \qquad \alpha_{i} \in \{0.1, 0.2, \dots, 2.0\},$$

where *s* corresponds to the contaminant (ranging from 1 to 35). The denominator c_s^{max} represents the maximum average monthly value of pollutant concentration *s* without any perturbations, calculated at the coordinates $(a_s^{i_{max}}, b_s^{j_{max}})$, and i_{max}, j_{max} are the grid indices of that point. The numerator represents the concentration value of the pollutant of interest for a specific set of perturbation parameter values $\alpha_i \in 0.1, \ldots, 2.0$, calculated at the point $(a_s^{i_{max}}, b_s^{j_{max}})$. Thus, the input data consist of pollutant concentrations normalized with respect to the maximum monthly mean value.

Before proceeding to the calculation of the Global Sensitivity Indices (GSIs) using Sobol's method, an approximation is performed.

During the UNI-DEM calculations, tables of model function values are generated. These tables depict the relationship between ozone concentration values at fixed perturbation parameter values $\alpha_i \in 0.1, ..., 2.0$, calculated at the point where the averaged maximum concentration is reached, and the corresponding averaged maximum for $\alpha = (1, ..., 1)$. Because the sensitivity analysis assumes that the model is represented by a function as defined in Equation (1), the first step involves using an approximation technique to create a continuous function with analytically specified properties.

The approximation step using polynomials of different degrees was investigated. We utilize second-degree polynomials, characterized by 28 unknown coefficients, as a means of approximation. These polynomials, denoted as $p_s^{(k)}(\mathbf{x})$, are employed to estimate the mesh function associated with the *s*-th chemical species :

$$p_{s}^{(k)}(\mathbf{x}) = \sum_{j=0}^{n} \sum_{(\nu_{1},\nu_{2},\dots,\nu_{d})\in N_{j}^{k}} a_{\nu_{1}\dots\nu_{d}} x_{1}^{\nu_{1}} x_{2}^{\nu_{2}}\dots x_{d}^{\nu_{d}}, \qquad k=2,$$

where $N_{j}^{k} = \left\{ (\nu_{1},\nu_{2},\dots,\nu_{d}) \mid \nu_{i}=0,1,\dots,k, \sum_{i=1}^{d} \nu_{i}=j \right\}.$

To evaluate the precision of the approximation, we employ the squared 2-vector norm. This norm, denoted as $|| p_s - r_s ||_2^2 = \sum_{l=1}^n [p_s(\mathbf{x}_l) - r_s(\mathbf{x}_l)]^2$, is computed as the sum of squared differences between the values of the polynomial p_s evaluated at specific points \mathbf{x}_l and the corresponding table values $r_s(x_l)$. Here, x_l belongs to the interval $[0.6; 1.4]^6$, and $r_s(x_l)$, l = 1, ..., n represents the table values obtained from running the UNI-DEM model.

To examine the impact of different rate constants of chemical reactions on air pollution concentrations, a numerical investigation was conducted in [48]. Specifically, only one input value of the model was altered while keeping the others fixed at 1.0. The results analysis, focusing on the reactions considered by the CBM-IV scheme, revealed the following observations: the reaction rates #1, 3, 22 exert a highly significant influence on the concentrations of thee ozone, whereas the impact of reaction rates and #7, 27, although less pronounced, remain significant. Conversely, the influence of reaction rate #28 can be disregarded. The investigation of the impact of changes in chemical rates on ozone concentrations for Genova on July 1998 is given in Figure 10.



Figure 10. Investigation of the impact of changes in chemical rates on ozone concentrations (Genova, July 1998).

For the SA regarding input emissions, our focus is on examining the impact of perturbing anthropogenic emissions, given as input data, on the UNI-DEM output. Specifically, we study the sensitivity of monthly average ammonia concentrations in relation to these emissions perturbations.

The input data themselves consist of four different components $\mathbf{E} = (\mathbf{E}^{\mathbf{A}}, \mathbf{E}^{\mathbf{N}}, \mathbf{E}^{\mathbf{S}}, \mathbf{E}^{\mathbf{C}})$:

| $\mathbf{E}^{\mathbf{A}}$ – ammonia (<i>NH</i> ₃); | $\mathbf{E}^{\mathbf{S}}$ – sulfur dioxide (SO ₂); |
|---|--|
| $\mathbf{E}^{\mathbf{N}}$ – nitrogen oxides (<i>NO</i> + <i>NO</i> ₂); | E ^C – anthropogenic hydrocarbons |

Similar to how chemical reaction rate constants are determined, the initial step of the calculations involves generating the necessary input data for conducting the sensitivity analysis (SA). In our specific case, this entails conducting a series of experiments using UNI-DEM and introducing specific perturbations to the emission data.

The outputs commonly used in UNI-DEM are the monthly average concentrations of various dangerous chemical species (or groups of species, depending on the specific chemical scheme) calculated at the grid points within the simulation area. Our focus for the SA is on the following chemical pollutants :

 s_1 —ozone (O_3),

 s_2 —ammonia (NH_3),

 s_3 —ammonium sulfate and ammonium nitrate ($NH_4SO_4 + NH_4NO_3$).

In fact, UNI-DEM produces aggregated concentrations of ammonium sulphate and ammonium nitrate. Therefore, the latter sum of pollutants is considered to be one aggregated pollutant in our further study.

Regarding the grid points of the computational domain three European cities with different climates and with different climatic conditions and pollution levels are selected in [47,49]: (i) Milan, (ii) Manchester, and (iii) Edinburgh.

Results from a large number of UNI-DEM operations with the following reduced emissions $E = (\alpha_1 E^N, \alpha_2 E^C, \alpha_3 E^S, \alpha_4 E^A)$ are needed for this SA study. The dedicated version of UNI-DEM is used in [47] to perform the necessary calculations for a set of different values of $\alpha = (\alpha_1, \alpha_2, \alpha_3, \alpha_4)$ in the area under the (in our case, the four-dimensional hypercubic region $[0.1, 1]^4$ and its subregion $[0.5, 1]^4$).

Following the parallel computations described in [47], a total of 15 tables were generated. Each table corresponds to a specific edge in the hypercube and contains model function values for reduced emissions adjacent to the (1, 1, 1, 1) vertices. These tables consist of nine columns, representing the results for each of the three pollutants, denoted as s_i in the selected three cities. The cities are identified by their closest grid point coordinates (a_i, b_i) . Within each column, the ratios $c_s^{\alpha}(a_i, b_i)/c_s(a_i, b_i)$ are presented. These ratios represent the average monthly concentration of pollutant *s* for a given set of parameter values α , uniformly distributed over the corresponding edge of the hypercube and divided by the corresponding concentration value for the baseline scenario where $\alpha = (1, 1, 1, 1, 1)$. Notably, all the data in the tables are normalized with respect to the baseline scenario, hence the presence of the value 1 as the first entry in each column (corresponding to $\alpha = (1, 1, 1, 1, 1)$). These tables serve as the basis for defining nine mesh functions (pertaining to different pollutants and locations) at various points within the hypercubic region, determined by the different values of α . The mesh functions derived from these tables are subsequently utilized as input data for the subsequent stage of the study [47].

The stage of approximation plays a vital role in bridging the gap between generating experimental data and applying mathematical techniques for sensitivity analysis. The accuracy of the resulting sensitivity indices greatly depends on the precise approximation of the data. Therefore, it is crucial to explore and identify suitable approximation tools for the table function.

We employ second-degree polynomials as a means of approximation, following the methodology described in [50]. Specifically, for the *s*-th chemical species, we utilize the polynomial $p_s(\mathbf{x})^{(k)}$ to approximate the values provided in the corresponding table. The polynomial takes the form:

$$p_s^{(k)}(\mathbf{x}) = \sum_{j=0}^k \sum_{\substack{\nu_1, \nu_2, \dots, \nu_d = 0 \\ \nu_1 + \dots + \nu_d = j}}^k a_{\nu_1 \dots \nu_d} x_1^{\nu_1} x_2^{\nu_2} \dots x_d^{\nu_d}, \qquad k = 2.$$

To assess the accuracy of the approximation, we utilize the squared 2-vector norm. This norm is computed as: $|| p_s - r_s ||_2^2 = \sum_{l=1}^n [p_s(\mathbf{x}_l) - r_s(\mathbf{x}_l)]^2$, where $\mathbf{x}_l \in [0.5; 1]^4$, and $r_s(\mathbf{x}_l)$, l = 1, ..., n represents the corresponding values obtained from the table through the execution of UNI-DEM.

It has been demonstrated that using higher degrees of approximating polynomials introduces more degrees of freedom, resulting in a larger number of unknown coefficients to be determined. The computation of these coefficients involves minimizing a functional, typically the sum of squared differences between the values of the grid function and the values of the approximating polynomial. However, employing high-degree polynomials does not necessarily lead to improved accuracy. In fact, it can lead to inferior results, such as less accurate calculation of very small polynomial coefficients. Additionally, high-degree polynomials can exhibit excessive flexibility for real mesh functions, similar to the well-known Gibbs phenomenon [30], where increasing the polynomial order worsens

the approximation in the uniform norm. Considering these factors, the decision to use second-degree polynomials as the primary approximation tool in this study was based on the aforementioned reasons.

5. Methods and Algorithms

Consider the following multidimensional problem:

$$I(f) := I = \int_{U^d} f(\mathbf{x}) d\mathbf{x},$$
(12)

where $\mathbf{x} \equiv (x_1, \dots, x_d) \in U^d \subset \mathbb{R}^d$ and $f \in C(U^d)$.

The most widely used quasi Monte Carlo algorithm, namely the *Sobol* sequence [51–53] is defined by:

$$\mathbf{x}_k \in \overline{\sigma}_i^{(k)}, k = 0, 1, 2, \dots$$

where $\overline{\sigma}_i^{(k)}$, $i \ge 1$ are the set of permutations on every 2^k , k = 0, 1, 2, ... subsequent points of the van der Corput sequence [54], defined by $n = \sum_{i=0}^{\infty} a_{i+1}(n)b^i$, $\phi_b(n) = \sum_{i=0}^{\infty} a_{i+1}(n)b^{-(i+1)}$ when b = 2. In binary, for the Sobol sequence we have that: $x_n^{(k)} = \bigoplus_{i\ge 0} a_{i+1}(n)v_i$, where v_i , i = 1, ..., s is the set of direction numbers [55].

The description of the modified Sobol sequences MCA-MSS-1, MCA-MSS-2, MCA-MSS-2S can be found in [56,57].

Until now, the best available modification of the Sobol sequence is the superconvergent Sobol–Burkardt method SOBOL-BURK based on the routines INSOBL and GOSOBL in ACM TOMS Algorithm 647 and ACM TOMS Algorithm 659 and Burkardt modification [58–62]. The original code can only compute the «next» element of the sequence. Our modification allows the user to specify the index of the desired element.

One of the best available methods for SA is also the DigitalSobol sequence DIGIT-SOBOL; this is a super-convergent digital sequence that is used for generating matrices based on an implementation of the Sobol sequence with 21,201 dimensions [63].

Now, to improve the Sobol sequence, we will define lattices.

Definition 10 ([64]). An N-point rank-one lattice rule in d dimensions is a quasi-Monte Carlo method with cubature points

$$x_k = \left(\left\{\frac{kz_1}{N}\right\}, \left\{\frac{kz_2}{N}\right\}, \dots, \left\{\frac{kz_d}{N}\right\}\right), \ k = 1, 2, \dots, N,$$
(13)

where $z \in \mathbb{Z}^s$ is known as the generating vector, $\mathbf{z} = (z_1, z_2, \dots z_d)$ and is an d dimensional integer vector having no common factors with N.

The corresponding quasi Monte Carlo approximation formula is given by:

$$Q(f) := \frac{1}{N} \sum_{k=1}^{N} f(x_k) = \frac{1}{N} \sum_{k=1}^{N} f\left(\left\{\frac{kz}{n}\right\}\right).$$
(14)

Definition 11 ([64]). For a given function class *F*, the worst-case error is defined as

$$e(Q,F) = \sup_{f \in F, \|f\| \le 1} |I(f) - Q(f)|.$$
(15)

In this study, we also construct two new super-convergent lattices based on component by component construction (CBC) method [65] of rank-one lattices with corresponding generating vectors with prime power of points with product weights LAT-PROD and with order-dependent weights LAT-ORDER. The worst-case error for the product weight lattice is given by

$$e^{2}(Q,K) = -1 + \frac{1}{n} \sum_{k=0}^{n-1} \prod_{j=1}^{d} (1 + \gamma_{j} w(x_{j}^{(k)})), \gamma_{u} = \prod_{j \in U} \gamma_{j}.$$
 (16)

and the worst-case error for the order-dependent weight lattice is given by

$$e^{2}(Q,K) = \frac{1}{n} \sum_{k=0}^{n-1} \sum_{l=1}^{d} \Gamma_{l} \sum_{u \subseteq D_{d}, |u|=l} \prod_{j \in U} w(x_{j}^{(k)})), \gamma_{u} = \Gamma_{|u|}.$$
(17)

It is proven in [66] that CBC method achieves optimal rate of convergence in weighted Korobov space $O(n^{-\alpha/2+\delta})$ and optimal rate of convergence in weighted Sobolev space $O(n^{-1+\delta})$ for $\delta > 0$ for the corresponding product weight and order-dependent weight lattice. This explain the fact that the constructed lattice outperforms the modified Sobol sequences.

6. Numerical Results

In this section, we will present some numerical results concerning UNI-DEM' SIs for the chosen European cities.

Table 1 contains the first-, second-, and total-order SIs of the considered model inputs. The results on the first- and second-order SIs of the ozone in Milan, Genova, Manchester, and Edinburgh, for July 1998, are represented graphically in Figure 11.

| Table 1. Sensitivity | y indices of input | parameters (for | r ozone concentrations) |
|----------------------|--------------------|-----------------|-------------------------|
|----------------------|--------------------|-----------------|-------------------------|

| Town | Genova | Milan | Manchester | Edinburgh |
|------------------------|-----------------------|---------------------|------------|-----------|
| f0 | 0.26588 | 0.26566 | 0.26526 | 0.26616 |
| D | 0.00249 | 0.00256 | 0.00245 | 0.00136 |
| <i>S</i> ₁ | 0.35858 | 0.36281 | 0.37165 | 0.33487 |
| S_2 | 0.29485 | 0.29936 | 0.26509 | 0.23399 |
| S_3 | 0.04652 | 0.04129 | 0.00997 | 0.05559 |
| S_4 | 0.26462 | 0.26276 | 0.32358 | 0.30133 |
| S_5 | 4.34×10^{-7} | $1.8 	imes 10^{-7}$ | 0.00023 | 0.00009 |
| <i>S</i> ₆ | 0.01904 | 0.01703 | 0.00857 | 0.04653 |
| $\sum_{i=1}^{6} S_i$ | 0.98361 | 0.98325 | 0.97909 | 0.97241 |
| S ₁₂ | 0.00556 | 0.00574 | 0.00568 | 0.00457 |
| S ₁₃ | 0.00048 | 0.00049 | 0.00024 | 0.00106 |
| S_{14} | 0.00516 | 0.00563 | 0.00809 | 0.00837 |
| S ₁₆ | 0.00031 | 0.00025 | 0.00018 | 0.00104 |
| S ₂₃ | 0.00038 | 0.00033 | 0.00005 | 0.00075 |
| S ₂₄ | 0.00349 | 0.00343 | 0.00516 | 0.00457 |
| S ₃₄ | 0.00045 | 0.00040 | 0.00015 | 0.00068 |
| S ₃₆ | 0.00016 | 0.00014 | 0.00039 | 0.00435 |
| $\sum_{i=1}^{6} S_i$ | 0.01639 | 0.01675 | 0.02092 | 0.02759 |
| S_1^{tot} | 0.37009 | 0.37493 | 0.38599 | 0.34993 |
| S_2^{fot} | 0.30442 | 0.30897 | 0.27625 | 0.24471 |
| $S_3^{\overline{t}ot}$ | 0.04799 | 0.04267 | 0.01098 | 0.06274 |
| S_{A}^{tot} | 0.27391 | 0.27239 | 0.33719 | 0.31559 |
| S_5^{fot} | 0.00015 | 0.00013 | 0.00089 | 0.00091 |
| $S_6^{\check{t}ot}$ | 0.01983 | 0.01766 | 0.00963 | 0.05371 |



Figure 11. Pie charts representation of first- and second-order sensitivity indices of the ozone in Milan, Genova, Manchester, and Edinburgh.

The results for the first- and second-order SIs for ammonia, ozone, and ammonium sulphate and ammonium nitrate in Milan, Manchester, and Edinburgh are presented numerically in Table 2.

The graphical representation in Figure 12 displays the findings regarding the first- and second-order SIs of ozone in Milan, Manchester, and Edinburgh during January 1997.



Figure 12. Pie chart representations of first- and second-order sensitivity indices of the ozone in Milan, Manchester, and Edinburgh.

They were obtained by applying VBM and in particular correlated sampling to compute all possible sensitivity measures to study the influence of four selected groups of air pollutant emissions over the concentration of the three important air pollutants mentioned above.

The described above advanced stochastic algorithms are applied to sensitivity studies with respect to input emission levels (SSIEL) and in accordance to some chemical reactions rates (SSCRR) of the concentration variations of UNI-DEM pollutants [47,48]. We denote the estimated quantity with EQ, the reference value with RF, the relative error with RE, and the approximate evaluation with AE.

| Pollutant | | NH ₃ | | | O ₃ | | NH ₄ S | $50_4 + NH_4$ | NO ₃ |
|---|--------------------|--------------------|--------------------|--------------------|-----------------------|--------------------|--------------------|--------------------|--------------------|
| Town | Milan | Manch. | Edinb. | Milan | Manch. | Edinb. | Milan | Manch. | Edinb. |
| f_0 | 0.048 | 0.049 | 0.049 | 0.059 | 0.068 | 0.062 | 0.044 | 0.045 | 0.044 |
| D | $2 	imes 10^{-4}$ | $4 	imes 10^{-4}$ | 3×10^{-4} | $1 	imes 10^{-5}$ | $5 	imes 10^{-5}$ | $4 	imes 10^{-6}$ | $4 	imes 10^{-4}$ | $4 	imes 10^{-5}$ | $4 	imes 10^{-5}$ |
| S_1 | 0.889 | 0.812 | 0.845 | $2 	imes 10^{-6}$ | $6 	imes 10^{-6}$ | $3 	imes 10^{-5}$ | 0.152 | 0.393 | 0.295 |
| S_2 | $2 	imes 10^{-4}$ | $1 	imes 10^{-4}$ | $4 	imes 10^{-4}$ | 0.156 | 0.791 | 0.387 | 0.017 | 0.006 | 0.012 |
| S_3 | 0.109 | 0.181 | 0.148 | 2×10^{-6} | 6×10^{-6} | 3×10^{-5} | 0.818 | 0.575 | 0.647 |
| S_4 | $4 	imes 10^{-5}$ | $8 	imes 10^{-5}$ | 3×10^{-4} | 0.826 | 0.209 | 0.589 | 0.002 | 0.002 | 0.008 |
| $\sum_{i=1}^{4} S_i$ | 0.999 | 0.994 | 0.994 | 0.983 | 0.999 | 0.976 | 0.991 | 0.976 | 0.962 |
| S ₁₂ | 2×10^{-5} | 5×10^{-6} | 1×10^{-5} | 7×10^{-7} | 1×10^{-7} | 8×10^{-6} | 0.001 | $2 	imes 10^{-4}$ | 4×10^{-4} |
| S ₁₃ | 0.001 | 0.006 | 0.006 | 7×10^{-7} | 1×10^{-7} | 8×10^{-6} | 0.007 | 0.024 | 0.033 |
| S_{14} | 2×10^{-6} | $7 	imes 10^{-6}$ | 5×10^{-5} | 7×10^{-7} | 1×10^{-7} | $8 	imes 10^{-6}$ | $1 	imes 10^{-4}$ | $1 	imes 10^{-4}$ | 0.001 |
| S ₂₃ | 5×10^{-6} | $7 	imes 10^{-8}$ | 5×10^{-6} | 7×10^{-7} | 1×10^{-7} | 8×10^{-6} | 3×10^{-4} | 3×10^{-5} | $6 	imes 10^{-5}$ |
| S_{24} | 5×10^{-6} | 2×10^{-6} | 1×10^{-4} | 0.017 | 3×10^{-4} | 0.024 | $4 	imes 10^{-4}$ | $1 	imes 10^{-4}$ | 0.004 |
| S_{34} | $7 	imes 10^{-7}$ | 3×10^{-6} | $2 	imes 10^{-5}$ | $7 	imes 10^{-7}$ | $1 	imes 10^{-7}$ | $8 	imes 10^{-6}$ | $2 	imes 10^{-5}$ | $5 	imes 10^{-5}$ | 3×10^{-4} |
| $\boxed{\sum_{i,j=1,i\leq j}^4 S_{ij}}$ | 0.001 | 0.006 | 0.006 | 0.017 | 3×10^{-4} | 0.024 | 0.009 | 0.024 | 0.038 |
| S ₁ ^{tot} | 0.891 | 0.818 | 0.851 | $4 	imes 10^{-6}$ | 6×10^{-6} | $6 	imes 10^{-5}$ | 0.161 | 0.417 | 0.329 |
| S_2^{tot} | $2 	imes 10^{-4}$ | $1 	imes 10^{-4}$ | $6 	imes 10^{-4}$ | 0.174 | 0.791 | 0.411 | 0.019 | 0.006 | 0.016 |
| $S_3^{\overline{t}ot}$ | 0.110 | 0.188 | 0.154 | $4 	imes 10^{-6}$ | $6 	imes 10^{-6}$ | $6 	imes 10^{-5}$ | 0.826 | 0.598 | 0.679 |
| S_4^{tot} | $5 	imes 10^{-5}$ | $9 	imes 10^{-5}$ | $5 	imes 10^{-4}$ | 0.844 | 0.209 | 0.613 | 0.003 | 0.003 | 0.013 |

Table 2. First -order , higher-order, and total sensitivity indices of input parameters.

For the SSIEL, we will investigate SA of the model output (in terms of mean monthly concentrations of several important pollutants—in our case, this is ammonia in Milan) in accordance with the perturbation of input emissions defined in the previous section.

For SSIEL, the results for REs for the AE of the f_0 , **D**, S_i , and S_i^{tot} are shown in Table 3, where the quantities are represented by eight-dimensional integrals.

In the case of the SSCRR, we will investigate the ozone concentration in Genova according to the rate variation of these chemical reactions: ## 1, 3, 7, 22 (time-dependent) and 27, 28 (time-independent) of the CBM-IV scheme [8].

In the case of the SSCRR, the results for REs for the AE of the f_0 , **D**, S_i , S_{ij} and S_i^{tot} , using the stochastic algorithms, are shown in Table 4, where the quantities are represented by 12-dimensional integrals.

Table 3. RE for AE of SIs $(n = 2^{16})$.

| EQ | RV | SOBOL- SEQ | MCA- MSS-1 | MCA- MSS-2 | MCA- MSS-2-S | DIGIT- Sobol | SOBOL- BURK | LAT- PROD | LAT- ORDER |
|---|---|--|---|--|--|--|---|--|---|
| S_1 S_2 S_3 S_4 | $\begin{array}{l} 9 \ \times \ 10^{-1} \\ 2 \ \times \ 10^{-4} \\ 1 \ \times \ 10^{-1} \\ 4 \ \times \ 10^{-5} \end{array}$ | 8×10^{-5} 3×10^{-2} 8×10^{-4} 7×10^{-2} | $\begin{array}{c} 8 \times 10^{-5} \\ 3 \times 10^{-2} \\ 8 \times 10^{-4} \\ 7 \times 10^{-2} \end{array}$ | 6×10^{-6} 4×10^{-3} 7×10^{-5} 1×10^{-2} | 5×10^{-4} 7×10^{-2} 1×10^{-2} 6×10^{-1} | 4×10^{-7} 4×10^{-6} 2×10^{-6} 2×10^{-4} | 3×10^{-7} 1×10^{-6} 3×10^{-6} 5×10^{-4} | 5×10^{-7} 1×10^{-5} 6×10^{-7} 9×10^{-3} | 5×10^{-8} 1×10^{-4} 6×10^{-7} 2×10^{-4} |
| $\begin{array}{c}S_1^{tot}\\S_2^{tot}\\S_3^{tot}\\S_4^{tot}\end{array}$ | $9 \times 10^{-1} \\ 2 \times 10^{-4} \\ 1 \times 10^{-1} \\ 5 \times 10^{-5}$ | $8 \times 10^{-5} 5 \times 10^{-3} 7 \times 10^{-4} 6 \times 10^{-2}$ | $8 \times 10^{-5} 2 \times 10^{-3} 7 \times 10^{-4} 6 \times 10^{-2}$ | $1 \times 10^{-5} \\ 1 \times 10^{-3} \\ 4 \times 10^{-5} \\ 1 \times 10^{-2}$ | $1 \times 10^{-3} \\ 3 \times 10^{-3} \\ 4 \times 10^{-3} \\ 1 \times 10^{-1}$ | $3 \times 10^{-7} \\ 1 \times 10^{-5} \\ 3 \times 10^{-6} \\ 4 \times 10^{-4}$ | $\begin{array}{c} 4 \times 10^{-7} \\ 4 \times 10^{-6} \\ 3 \times 10^{-6} \\ 2 \times 10^{-4} \end{array}$ | $5 \times 10^{-7} \\ 6 \times 10^{-6} \\ 7 \times 10^{-7} \\ 7 \times 10^{-3}$ | $\begin{array}{c} 4 \times 10^{-8} \\ 9 \times 10^{-5} \\ 7 \times 10^{-7} \\ 1 \times 10^{-4} \end{array}$ |

| EQ | RV | SOBOL- SEQ | MCA- MSS-1 | MCA- MSS-2 | MCA- MSS-2-S | DIGIT- Sobol | SOBOL- BURK | LAT- PROD | LAT- ORDER |
|-------------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|----------------------|
| S_1 | $4 	imes 10^{-1}$ | 1×10^{-4} | $4 	imes 10^{-4}$ | $2 	imes 10^{-4}$ | 2×10^{-2} | 5×10^{-6} | 5×10^{-5} | 6×10^{-5} | 8×10^{-7} |
| S_2 | 3×10^{-1} | 3×10^{-5} | $2 	imes 10^{-4}$ | 3×10^{-4} | 6×10^{-2} | 6×10^{-6} | 5×10^{-5} | $6 	imes 10^{-5}$ | $6 	imes 10^{-6}$ |
| S_3 | 5×10^{-2} | $2 	imes 10^{-4}$ | 2×10^{-3} | $9 	imes 10^{-4}$ | $8 	imes 10^{-2}$ | 1×10^{-4} | 9×10^{-5} | 2×10^{-3} | $6 	imes 10^{-5}$ |
| S_4 | 3×10^{-1} | 3×10^{-4} | 2×10^{-5} | $2 	imes 10^{-4}$ | $4 	imes 10^{-3}$ | 1×10^{-5} | 1×10^{-4} | $9 	imes 10^{-5}$ | $4 	imes 10^{-7}$ |
| S_5 | $4 	imes 10^{-7}$ | 3×10^{-1} | $7 	imes 10^{0}$ | $7 	imes 10^{-2}$ | 2×10^2 | 7×10^{-2} | 1×10^{-1} | $4 	imes 10^{-3}$ | 2×10^{-2} |
| S_6 | 2×10^{-2} | $3 	imes 10^{-4}$ | $1 	imes 10^{-3}$ | $3 	imes 10^{-4}$ | $4 	imes 10^{-2}$ | $4 	imes 10^{-6}$ | 3×10^{-3} | $1 	imes 10^{-5}$ | $2 	imes 10^{-5}$ |
| S ₁ ^{tot} | 4×10^{-1} | 1×10^{-4} | 4×10^{-5} | 2×10^{-4} | 5×10^{-2} | $8 	imes 10^{-7}$ | 5×10^{-5} | 6×10^{-5} | 1 × 10 ⁻⁶ |
| S_2^{tot} | 3×10^{-1} | $4 	imes 10^{-5}$ | 5×10^{-4} | $2 	imes 10^{-4}$ | 3×10^{-2} | 3×10^{-6} | 5×10^{-5} | 1×10^{-3} | 5×10^{-5} |
| $S_3^{\overline{t}ot}$ | 5×10^{-2} | 3×10^{-4} | 2×10^{-3} | $8 	imes 10^{-4}$ | 4×10^{-2} | 1×10^{-4} | $4 	imes 10^{-5}$ | 1×10^{-3} | 5×10^{-5} |
| S_4^{tot} | 3×10^{-1} | $2 	imes 10^{-4}$ | 5×10^{-4} | $2 	imes 10^{-4}$ | $4 	imes 10^{-2}$ | 2×10^{-5} | 3×10^{-4} | $1 	imes 10^{-4}$ | 1×10^{-6} |
| S_5^{tot} | $2 	imes 10^{-4}$ | 7×10^{-3} | 1×10^{-2} | $4 	imes 10^{-3}$ | 1×10^{0} | 9×10^{-3} | $6 	imes 10^{-4}$ | $1 	imes 10^{-4}$ | $6 	imes 10^{-4}$ |
| S_6^{tot} | 2×10^{-2} | $4 	imes 10^{-4}$ | 1×10^{-3} | 3×10^{-4} | 4×10^{-2} | 1×10^{-5} | 5×10^{-5} | 2×10^{-5} | 3×10^{-5} |
| S_{12} | $6 	imes 10^{-3}$ | $2 	imes 10^{-4}$ | 5×10^{-3} | 1×10^{-3} | $7 	imes 10^{-1}$ | 1×10^{-4} | 2×10^{-4} | $8 	imes 10^{-5}$ | $6 	imes 10^{-6}$ |
| S_{14} | 5×10^{-3} | 2×10^{-3} | 2×10^{-2} | 2×10^{-3} | 1×10^{0} | 2×10^{-5} | 2×10^{-4} | $2 	imes 10^{-4}$ | 1×10^{-5} |
| S_{24} | 3×10^{-3} | 1×10^{-3} | 2×10^{-2} | 6×10^{-3} | 1×10^{0} | $2 	imes 10^{-4}$ | 9×10^{-5} | $2 	imes 10^{-4}$ | 3×10^{-5} |
| S_{45} | $1 	imes 10^{-5}$ | $4 	imes 10^{-2}$ | $1 	imes 10^{-1}$ | 2×10^{-2} | 4×10^{0} | 1×10^{-2} | 3×10^{-3} | 5×10^{-4} | $4 	imes 10^{-3}$ |

Table 4. RE for AE of SIs $(n = 2^{16})$.

7. Discussion and Applicability

We could make the following comments about the chemical reaction rates:

- It can be expected that the values of the higher-order SIs are relatively small and close to zero, given that the values of the SIs of the TSIs are close to each other. UNI-DEM mathematical model is additive based on the selected input parameters, specifically, the rates of chemical reactions.
- A new important input parameter, the rate of the time-dependent chemical reaction #1, has been identified.
- The findings of this study align completely with the conclusions drawn in a previous study regarding the significance of model inputs.

The following comments can be made in the case of input emissions regarding the different cities.

On ammonia concentrations:

The most influential pollutant emissions are the ammonia emissions themselves, accounting for 81–89% of the impact. Sulfur dioxide emissions also have a notable influence (11–18%), with the largest impact observed in the Manchester area. This pattern is consistent across all three cities examined, although the influence in the southernmost city, Milan, is slightly higher. Higher-order effects are almost negligible, except for a joint effect (0.1–0.6%) of the two aforementioned groups of air pollutants. Total effects primarily consist of the corresponding main effects, but in Manchester and Edinburgh, there is a slight contribution from the joint effect of ammonia and sulfur dioxide emissions.

• On ozone concentrations:

•

The most influential pollutant emissions in Milan and Edinburgh are anthropogenic hydrocarbons (59–83%), whereas nitrogen oxides play a dominant role (79%) in Manchester. In the areas of Milan and Edinburgh, nitrogen oxides emissions (16–39%) have a significant influence. The impact of nitrogen oxides emissions and anthropogenic hydrocarbons is relatively balanced in Edinburgh. Second-order interaction effects in Manchester are almost negligible (even $S_{24} \ll 0.1\%$), whereas in Milan and Edinburgh, the joint effect accounts for approximately 2%. Total effects are primarily driven by the corresponding main effects, but Milan and Edinburgh also exhibit a slight contribution from the joint effects of anthropogenic hydrocarbons and nitrogen oxides emissions. On ammonium sulfate and ammonium nitrate concentrations:

The most influential pollutant emissions are sulfur dioxide emissions (58–82%), with ammonia emissions also having a significant but smaller impact (15–39%). In Manchester, the influence of ammonia and sulfur dioxide emissions is comparatively bal-

anced. All four groups of pollutants have an influence on the considered important species, with nitrogen oxides and anthropogenic hydrocarbons exhibiting a slight but not negligible effect. Second-order effects in the Manchester area are mostly negligible, except for $S_{13} \approx 2.5\%$. In Edinburgh, three second-order interaction effects contribute to the corresponding total effects (0.1–3.3%).

The following observations for SIs can be made regarding the SSIEL:

- From Table 3, it can be deduced that the most effective algorithms of all of the first-order SIs and TSIs, the most accurate is the order-dependent weight lattice, except for S₂ and S₂^{tot}, where the Sobol–Burkardt algorithm produces better results.
- The product weight lattice is generally worse than the order-dependent lattice, but it produce the same relative error for S_3 and S_3^{tot} .
- In [67], it is emphasized that having the smallest possible SIs is crucial for the model. In our case, these smallest SIs are S_4 and S_4^{tot} , and the order-dependent weight lattice outperform the other sequences for these SIs, but for S_4 the digital Sobol algorithm produced the same relative error as the order-dependent weight lattice.
- Generally, in the case of SSIEL, our lattice sequences yield significantly better results than the original Sobol sequence and its modifications, namely MCA-MSS-1, 2, and 2S. Similarly, in the case of the SSCRR, the following can be observed:
- For a sample size of $N = 2^{16}$ in Table 4, order-dependent weight lattice produces the best results for most of the cases, except S_5 , S_6 , S_1^{tot} , S_2^{tot} , S_3^{tot} , S_6^{tot} , and S_{45} .
- As mentioned earlier, having the smallest possible SIs is crucial for the model. In this case, these smallest SIs are S₅, S₄₅, and S₅^{tot}, and the product weight lattice produce better results than the other algorithms for these SIs, only in the case of S₅^{tot} Sobol–Burkardt produce the same result as product weight lattice.
- The digital Sobol sequence is the most accurate for S_6 , S_1^{tot} , and S_2^{tot} , whereas Sobol–Burkardt is the most accurate for S_3^{tot} .
- Generally, the order-dependent weighted and product weighted lattice significantly outperforms the original Sobol sequence and its modifications, MCA-MSS-1, -2, and -2S.

In conclusion, the two developed lattices are the most effective approaches among the benchmarked methods, as indicated by the relative error values. For some of the SIs, the most well-known Sobol algorithms, the digital Sobol and Sobol–Burkardt algorithms, produce slightly better results, but this is not the case for the smallest in value SIs. When applied to multidimensional air pollution sensitivity analysis, these sequences demonstrate superiority over the majority of existing methods. It should be noted that the lattice with a generating vector consisting of a prime number of points and with product weights produce the best results for the smallest in value SIs for the second case of SSCRR, whereas the lattice with the generating vector, a prime number of points, and order-dependent weights produces the most accurate results for the smallest in value SIs for the first case of SSIEL.

8. Conclusions

This study focuses on the application of a sophisticated digital twin named DIGITAL AIR to examine the problem of high air pollution levels in different regions of Europe. The study employs a range of tools and techniques to effectively investigate this issue.

The UNI-DEM mathematical model plays a central role in this study, requiring the implementation of highly efficient and accurate numerical algorithms. These algorithms are executed on state-of-the-art supercomputers to ensure reliable and precise simulations of air pollution dynamics.

To support the modeling efforts, extensive datasets are utilized, encompassing meteorological data, emission data, and geographical information. These datasets provide essential input parameters for the simulations and contribute to the overall accuracy and reliability of the findings. To account for future climate changes and their impact on air pollution, the study incorporates carefully designed climatic scenarios. These scenarios consider the anticipated increase in temperatures and the corresponding changes in natural (biogenic) emissions. By incorporating these future projections, the study aims to provide insights into the potential long-term effects of climate change on air quality.

Visualizing the obtained numerical results is an integral part of the study, and graphical programs are employed for this purpose. These programs enable the researchers to analyze and interpret the simulation outcomes in a visually accessible manner, facilitating a comprehensive understanding of the complex air pollution patterns.

Additionally, this study introduces a further exploration of DIGITAL AIR through a multidimensional sensitivity analysis conducted using advanced stochastic methods based on superconvergent lattice sequences. This analysis aims to investigate the sensitivity of the model and its outputs to variations in input parameters and uncertainties. By employing stochastic techniques, the study can account for the inherent randomness and variability in the system, leading to a more comprehensive understanding of the model's behavior and its robustness in different scenarios.

In summary, this research utilizes the DIGITAL AIR digital twin and employs a range of tools, including the UNI-DEM mathematical model, extensive datasets, carefully designed climatic scenarios, and graphical programs for visualization. The investigation is extended through a sensitivity analysis using advanced stochastic methods based on powerful lattice and digital sequences, contributing to a more comprehensive understanding of air pollution dynamics and the potential impacts of climate change.

The current version of UNI-DEM, although a powerful mathematical model for air pollution analysis, has certain limitations that should be acknowledged. One notable limitation is that it does not account for PM10 (consists of small particles suspended in the air, such as dust, pollen, soot, and other solid or liquid pollutants) in its calculations. These particles are small enough to be inhaled into the respiratory system, posing potential health risks. Monitoring and controlling PM10 levels is crucial for assessing air quality and understanding its impact on human health and the environment. However, future iterations of UNI-DEM are expected to address this limitation by incorporating PM10 data and considering their impact on air pollution dynamics. By taking into account this important particulate matter, the model will provide a more comprehensive and accurate representation of air quality, contributing to a deeper understanding of the factors influencing air pollution levels.

In future research, a more comprehensive comparison will be conducted between the newly developed lattice approach and the most advanced digital sequences. The aim will be to explore and evaluate the performance of these methods in greater detail. Moreover, the results of our study presented in this paper through sensitivity analysis will have a multifaceted and highly significant impact. By utilizing the insights gained from our sensitivity analysis, the mathematical model will provide a more precise assessment of agricultural losses. Additionally, it will serve a crucial role in estimating the detrimental effects of emissions on human health.

Author Contributions: Conceptualization, V.T. and I.D.; methodology, V.T. and I.D.; software, V.T.; validation, V.T.; formal analysis, V.T.; investigation, V.T.; resources, V.T.; data curation, V.T.; writing—original draft preparation, V.T.; writing—review and editing, V.T.; visualization, V.T; supervision, V.T. and I.D.; project administration, I.D.; funding acquisition, I.D. All authors have read and agreed to the published version of the manuscript.

Funding: The development of stochastic methods is supported by the Project BG05M2OP001-1.001-0004 UNITe, funded by the Operational Programme "Science and Education for Smart Growth", co-funded by the European Union trough the European Structural and Investment Funds. The methodology for the environmental investigation is supported by the Bulgarian National Science Fund (BNSF) under Project KP-06-N52/5 "Efficient methods for modeling, optimization and decision making". The sensitivity study of the Digital Twin is supported by BNSF under Bilateral Project KP-06-Russia/17 "New Highly Efficient Stochastic Simulation Methods and Applications".

The computational simulations are supported by the BNSF under Project KP-06-M62/1 "Numerical deterministic, stochastic, machine and deep learning methods with applications in computational, quantitative, algorithmic finance, biomathematics, ecology and algebra" from 2022.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

| IPCC | International Panel for Climate Change's |
|---------|--|
| EMEP | European Monitoring and Evaluation Programme |
| EU | European Union |
| UNI-DEM | Directory of open access journals |
| SA | Sensitivity analysis |
| VBC | Variance-based methods |
| MC | Monte Carlo |
| QMC | quasi Monte Carlo |
| OAT | "one at a time" |
| SI | sensitivity index |
| TSI | Total Sensitivity Index |
| GSIs | global sensitivity indices |
| HDMR | high dimensional model representation |
| ANOVA | analysis of variance |
| CBM-IV | Carbon Bond Mechanism |
| PDEs | partial differential Equations |
| ODEs | ordinary differential Equations |
| MCA-MSS | Monte Carlo algorithm based on modified Sobol sequences |
| SSIEL | sensitivity studies with respect to input emission levels |
| SSCRR | sensitivity studies with respect to chemical reactions rates |
| | |

References

- IPCC. Summary for Policymakers. In Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change; Masson-Delmotte, V. P., Zhai, A., Pirani, S.L., Connors, C., Péan, S., Berger, N., Caud, Y., Chen, L., Goldfarb, M.I., Gomis, M., et al., Eds.; Cambridge University Press: Cambridge, UK, 2021.
- Topping, D.; Bannan Thomas, J.; Coe, H.; Evans, J.; Jay, C.; Murabito, E.; Robinson, N. Digital Twins of Urban Air Quality: Opportunities and Challenges. *Front. Sustain. Cities* 2021, *3*, 141. Available online: https://www.frontiersin.org/articles/10.3389/ frsc.2021.786563 (accessed on 15 December 2021). [CrossRef]
- 3. Nativi, S.; Mazzetti, P.; Craglia, M. Digital Ecosystems for Developing Digital Twins of the Earth: The Destination Earth Case. *Remote Sens.* **2021**, *13*, 2119. [CrossRef]
- Pedersen, A.N.; Borup, M.; Brink-Kjær, A.; Christiansen, L.E.; Mikkelsen, P.S. Living and Prototyping Digital Twins for Urban Water Systems: Towards Multi-Purpose Value Creation Using Models and Sensors. *Water* 2021, 13, 592. [CrossRef]
- 5. Dembski, F.; Wössner, U.; Letzgus, M.; Ruddat, M.; Yamu, C. Urban Digital Twins for Smart Cities and Citizens: The Case Study of Herrenberg, Germany. *Sustainability* **2020**, *12*, 2307. [CrossRef]
- 6. Bauer, P.; Stevens, B.; Hazeleger, W. A digital twin of Earth for the green transition. Nat. Clim. Chang. 2021, 11, 80–83. [CrossRef]
- 7. Zlatev, Z.; Dimov, I. Using a digital twin to study the influence of climatic changes on high ozone levels in bulgaria and europe. *Atmosphere* **2022**, *13*, 932. [CrossRef]
- 8. Zlatev, Z. *Computer Treatment of Large Air Pollution Models*; KLUWER Academic Publishers: Dorsrecht, The Netherlands; Boston, MA, USA; London, UK, 1995.
- 9. Zlatev, Z.; Dimov, I.T.; Georgiev, K. Three-dimensional version of the Danish Eulerian model. *Z. Angew. Math. Mech.* **1996**, 76, 473–476.
- 10. Zlatev, Z.; Dimov, I. Computational and Numerical Challengies in Environmental Modelling; Elsevier: Amsterdam, The Netherlands, 2006.
- 11. Zlatev, Z. Impact of Future Climatic Changes on High Ozone Levels in European Suburban Areas. *Clim. Change* **2010**, *101*, 447–483. [CrossRef]

- Zlatev, Z.; Dimov, I.; Faragó, I.; Georgiev, K. Large-scale Air Pollution Modelling in Europe Under Different Climatic Scenarios . *Int. J. Big Data Min. Glob. Warm.* 2019, 1, 1950009. Available online: https://www.worldscientific.com/doi/abs/10.1142/S26305 34819500098 (accessed on 20 June 2023). [CrossRef]
- Zlatev, Z.; Dimov, I.; Georgiev, K. Relations between Climatic Changes and High Pollution Levels in Bulgaria. *Open J. Appl. Sci.* 2016, *6*, 386–401. [CrossRef]
- Zlatev, Z.; Faragó, I.; Havasi, Á. Impact of Climatic Changes on Pollution Levels. In *Mathematical Problems in Meteorological Modelling*; Bátkai, A., Csomós, P., Faragó, I., Horányi, A., Szépszó, G., Eds.; Springer Series on "Mathematics in Industry"; Springer: Berlin, Germany, 2016; Volume 24, pp. 129–161. [CrossRef]
- Zlatev, Z.; Georgiev, K.; Dimov, I. Influence of Climatic Changes on Pollution Levels in the Balkan Peninsula. *Comput. Math. Appl.* 2011, 65, 544–562. [CrossRef]
- 16. Zlatev, Z.; Havasi, Á.; Faragó, I. Influence of Climatic Changes on Pollution Levels in Hungary and Its Surrounding Countries. *Atmosphere* **2011**, *2*, 201–221. [CrossRef]
- 17. Zlatev, Z.; Moseholm, L. Impact of Climatic Changes on Pollution Levels in Denmark. Ecol. Model. 2008, 217, 305–319. [CrossRef]
- Zlatev, Z.; Syrakov, D. A Fine Resolution Modelling Study of Pollution Levels in Bulgaria. Part 1: SO₂ and NO_x Emissions. *Int. J. Environ. Pollut.* 2004, 22, 186–202. [CrossRef]
- Zlatev, Z.; Syrakov, D. A Fine Resolution Modelling Study of Pollution Levels in Bulgaria. Part 2: High Ozone Levels. *Int. J. Environ. Pollut.* 2004, 22, 203–222. [CrossRef]
- Kucherenko, S.; Song, S. Derivative-Based Global Sensitivity Measures and Their Link with Sobol' Sensitivity Indices. In *Monte Carlo and Quasi-Monte Carlo Methods*; Cools, R., Nuyens, D., Eds.; Springer Proceedings in Mathematics & Statistics; Springer: Cham, Switzerland, 2016; Volume 163.
- Sobol´, I.M.; Kucherenko, S.S. On global sensitivity analysis of quasi-Monte Carlo algorithms. *Monte Carlo Methods Appl.* 2005, 11, 83–92. [CrossRef]
- Kucherenko, S.; Rodriguez-Fernandez, M; Pantelides, C.; Shah, N. Monte Carlo evaluation of derivative-based global sensitivity measures. *Reliab. Eng. Syst. Saf.* 2009, 94, 1135–1148. [CrossRef]
- 23. Ferretti, F.; Saltelli, A.; Tarantola, S. Trends in sensitivity analysis practice in the last decade journal. *Sci. Total Environ. Spec. Issue Hum. Biota Expo.* **2016**, *568*, 666–670. [CrossRef]
- 24. Chan, K.; Saltelli, A.; Tarantola S. Sensitivity analysis of model output: Variance-based methods make the difference. In Proceedings of the 1997 Winter Simulation Conference, Atlanta, GA, USA, 7–10 December 1997; pp. 261–268.
- 25. Saltelli, A.; Chan K.; Scott M. Sensitivity Analysis; John Wiley & Sons Publishers: London, UK, 2000.
- 26. Saltelli, A.; Tarantola, S.; Chan K. A quantitative model-independent method for global sensitivity analysis of model output. Source. *Technometrics Arch.* **1999**, *41*, 39–56. [CrossRef]
- Homma, T.; Saltelli, A. Importance measures in global sensitivity analysis of nonlinear models. *Reliab. Eng. Syst. Saf.* 1996, 52, 1–17. [CrossRef]
- Sandewall, E. Combining logic and differential equations for describing real-world system. In Proceedings of the First International Conference on Principles of Knowledge Representation and Reasoning, Toronto, ON, Canada, 15–18 May 1989; Brachmann, R.J., Levesque, H., Reiter, R., Eds.; Morgan Kaufmann: Los Altos, CA, USA, 1989; pp. 412–320.
- 29. Saltelli, A.; Tarantola, S.; Campolongo, F.; Ratto, M. Sensitivity Analysis in Practice: A Guide to Assessing Scientific Models; Halsted Press: New York, NY, USA, 2004.
- Saltelli, A. Making best use of model valuations to compute sensitivity indices. Comput. Phys. Commun. 2002, 145, 280–297. [CrossRef]
- 31. Dimov I.T. Monte Carlo Methods For Applied Scientists; World Scientific: Hackensack, NJ, USA; London, UK; Singapore, 2007.
- Georgiev, S.G.; Vulkov, L.G. Computation of the unknown volatility from integral option price observations in jump–diffusion models. *Math. Comput. Simul.* 2021, 188, 591–608. [CrossRef]
- Kostadinova, V.; Georgiev, I.; Mihova, V.; Pavlov, V. An application of Markov chains in stock price prediction and risk portfolio optimization. AIP Conf. Proc. 2021, 2321, 030018.
- 34. Saltelli, A.; Ratto, M.; Andres, T.; Campolongo, F.; Cariboni, J.; Gatelli, D.; Saisana, M.; Tarantola, S. *Global Sensitivity Analysis: The Primer;* John Wiley & Sons, Ltd.: Hoboken, NJ, USA, 2008.
- 35. Morris, M. Factorial Sampling Plans for Preliminary Computational Experiments. Technometrics 1991, 33, 161–174. [CrossRef]
- 36. Sobol', I.M. Sensitivity estimates for nonlinear mathematical models. *Mat. Model.* **1990**, *2*, 112–118.
- 37. Sobol, I.M. Sensitivity estimates for nonlinear mathematical models. Math. Model. Comput. Exp. 1993, 4, 407–414.
- Jacques, J.; Lavergne, C.; Devictor, N. Sensitivity analysis in presence of modele uncertainty and correlated inputs. *Reliab. Eng.* Syst. 2006, 91, 1126–1134. [CrossRef]
- 39. Sobol', I.M. Theorem and Examples on High Dimensional Model Representation. *Reliab. Eng. Syst. Saf.* 2003, 79, 187–193. [CrossRef]
- 40. Rabitz, H.; Alis, O. Managing the tyranny of parameters in mathematical modelling. *Sensit. Anal.* 2000, 199–223.
- Sobol', I.M. Global sensitivity indices for nonlinear mathematical models and their Monte Carlo estimates. *Math. Comput. Simul.* 2001, 55, 271–280. [CrossRef]
- 42. Dimov, I.; Zlatev, Z. Testing the sensitivity of air pollution levels to variations of some chemical rate constants. *Notes Numer. Fluid Mech.* **1997**, *62*, 167–175.

- 43. Zlatev, Z.; Dimov, I.T.; Georgiev, K. Modeling the long-range transport of air pollutants. *IEEE Comput. Sci. Eng.* **1994**, *1*, 45–52. [CrossRef]
- 44. Houghton, J.T.; Ding, Y.; Griggs, D.J.; Noguer, M.; van der Linden, P.J.; Dai, X.; Maskell, K.; Johnson, C.A. (Eds.) *Climate Change* 2001: *The Scientific Basis*; Cambridge University Press: Cambridge, UK; New York, NY, USA; Melbourne, Australia; Madrid, Spain; Cape Town, South Africa, 2001.
- 45. Anastasi, C.; Hopkinson, L.; Simpson, V.J. Natural Hydrocarbon Emissions in the United Kingdom. *Atmos. Environ.* **1991**, 25*A*, 1403–1408. .. [CrossRef]
- Simpson, D.; Guenther, A.; Hewitt, C.N.; Steinbrecher, R. Biogenic Emissions in Europe: I. Estimates and Uncertainties. J. Geophys. Res. 1995, 100, 22875–22890. [CrossRef]
- 47. Dimov, I.T.; Georgieva, R.; Ostromsky, T.; Zlatev, Z. Sensitivity Studies of Pollutant Concentrations Calculated by UNI-DEM with Respect to the Input Emissions. *Cent. Eur. J. Math. Methods Large Scale Sci. Comput.* **2013**, *11*, 1531–1545. [CrossRef]
- 48. Dimov, I.T.; Georgieva, R.; Ostromsky, T.; Zlatev, Z. Variance-based Sensitivity Analysis of the Unified Danish Eulerian Model According to Variations of Chemical Rates. In Numerical Analysis and Its Applications, Proceedings of the 5th International Conference, NAA 2012, Lozenetz, Bulgaria, 15–20 June 2012; Dimov, I., Faragó, I., Vulkov, L., Eds.; Springer: Berlin, Germany, 2013; pp. 247–254.
- 49. Dimov, I.T.; Georgieva, R.; Ivanovska, S.; Ostromsky, T.; Zlatev, Z. Studying the sensitivity of pollutants' concentrations caused by variations of chemical rates. *J. Comput. Appl. Math.* **2010**, *235*, 391–402. [CrossRef]
- 50. Sudret, B. Global sensitivity analysis using polynomial chaos expansion. Reliab. Eng. Syst. Saf. 2008, 93, 964–979. [CrossRef]
- 51. Antonov, I.; Saleev, V. An Economic Method of Computing LP_{τ} -sequences. USSR Comput. Math. Phys. 1979, 19, 252–256. [CrossRef]
- 52. Bratley, P.; Fox, B. Algorithm 659: Implementing Sobol's Quasirandom Sequence Generator. *ACM Trans. Math. Softw.* **1988**, 14, 88–100. [CrossRef]
- Bratley, P.; Fox, B.; Niederreiter, H. Implementation and Tests of Low Discrepancy Sequences. ACM Trans. Model. Comput. Simul. 1992, 2, 195–213. [CrossRef]
- 54. van der Corput, J.G. Verteilungsfunktionen (Erste Mitteilung) (PDF). Proc. K. Akad. Van Wet. Amst. 1935, 38, 813–821. (In German)
- 55. Bratley, P.; Fox, B.; Schrage, L. A Guide to Simulation, 2nd ed.; Springer: Berlin, Germany, 1987. ISBN: 0387964673, LC : QA76.9.C65.B73.
- 56. Dimov, I.T.; Georgieva, R. Monte Carlo Method for Numerical Integration based on Sobol' Sequences. In *Numerical Methods and Applications*; Dimov, I., Dimova, S., Kolkovska, N., Eds.; LNCS 6046; Springer: Berlin, Germany, 2011; pp. 50–59.
- 57. Dimov, I.T.; Georgieva, R. Multidimensional Sensitivity Analysis of Large-scale Mathematical Models. In Numerical Solution of Partial Differential Equations: Theory, Algorithms, and Their Applications; Iliev, O.P., Margenov, S.D., Minev, P.D., Vassilevski, P.S., Zikatanov, L.T., Eds.; Springer Proceedings in Mathematics & Statistics 45; Springer Science+Business Media: New York, NY, USA, 2013; pp. 137–156.
- Fox, B. Algorithm 647: Implementation and Relative Efficiency of Quasirandom Sequence Generators. ACM Trans. Math. Softw. 1986, 12, 362–376. [CrossRef]
- Joe, S.; Kuo, F. Remark on Algorithm 659: Implementing Sobol's Quasirandom Sequence Generator. ACM Trans. Math. Softw. 2003, 29, 49–57. [CrossRef]
- 60. Niederreiter, H. Random Number Generation and quasi-Monte Carlo Methods; SIAM: Philadelphia, PA, USA, 1992; ISBN13: 978-0-898712-95-7, LC: QA298.N54.
- 61. Press, W.; Flannery, B.; Teukolsky, S.; Vetterling, W. *Numerical Recipes in FORTRAN: The Art of Scientific Computing*, 2nd ed.; Cambridge University Press: Cambridge, UK, 1992; ISBN: 0-521-43064-X, LC: QA297.N866.
- 62. Sobol, I. Uniformly Distributed Sequences with an Additional Uniform Property. USSR Comput. Math. Math. Phys. 1977, 16, 236–242. [CrossRef]
- Joe, S.; Kuo, F.Y. Constructing Sobol' sequences with better two-dimensional projections. SIAM J. Sci. Comput. 2008, 30, 2635–2654. [CrossRef]
- 64. Dick, J.; Pillichshammer, F. Digital Nets and Sequences: Discrepancy Theory and Quasi–Monte Carlo Integration; Cambridge University Press: Cambridge, UK, 2010.
- 65. Baldeaux, J.; Dick, J.; Leobacher, G.; Nuyens, D.; Pillichshammer, F. Efficient calculation of the worst-case error and (fast) component-by-component construction of higher order polynomial lattice rules. *Numer. Algorithms* **2012**, *59*, 403–431. [CrossRef]
- 66. Nuyens, D.; Cools, R. Fast algorithms for component-by-component construction of rank-1 lattice rules in shift-invariant reproducing kernel Hilbert spaces. *Math. Comput.* **2006**, *75*, 903–920. [CrossRef]
- Archer, G.; Saltelli, A.; Sobol, I. Sensitivity measures, ANOVA-like techniques and the use of bootstrap. *J. Stat. Comput. Simul.* 1997, 58, 99–120. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.